Author's response to reviews

Title: Effects of lifestyle education program for type 2 diabetes patients in clinics: a cluster randomized controlled trial

Authors:

Misa Adachi (adachi@nutrisupport.co.jp)
Kazue Yamaoka (kazue@med.teikyo-u.ac.jp)
Mariko Watanabe (mariko_w@swu.ac.jp)
Masako Nishikawa (mnishikawa@niph.go.jp)
Itsuro Kobayashi (doctor@m9.dion.ne.jp)
Eisuke Hida (e.hida.web@gmail.com)
Toshiro Tango (tango@medstat.jp)

Version: 2 Date: 30 November 2012

Author's response to reviews: see over